Author’s response to reviews

Title: Soluble Klotho is not independently associated with cardiovascular disease in a population of dialysis patients

Authors:

Maurits S Buiten (m.s.buiten@lumc.nl)
Mihály K de Bie (M.K.de_Bie@lumc.nl)
Annet Bouma-de Krijger (A.Bouma@vumc.nl)
Bastiaan van Dam (B.van.Dam@mca.nl)
Fiedo W Dekker (F.W.Dekker@lumc.nl)
J Wouter Jukema (J.W.Jukema@lumc.nl)
Ton J Rabelink (A.J.Rabelink@lumc.nl)
Joris I Rotmans (J.I.Rotmans@lumc.nl)

Version: 5
Date: 13 November 2014

Author’s response to reviews:

From: dr. J.I. Rotmans
Department of Nephrology
Leiden University Medical Center
P.O. Box 9600, 2300RC Leiden
The Netherlands
MS: 1938025955134764

Leiden, November 11th, 2014

Dear dr. Covic,

We hereby resubmit our manuscript entitled: ‘Soluble Klotho is not independently associated with cardiovascular disease in a population of dialysis patients’. We would like to thank you as well as the reviewers for their comments. We updated the ethics statement.

We hope that you will find the revised paper suitable for publication in “BMC Nephrology”

Sincerely,

Joris Rotmans, M.D., PhD.